Author's response to reviews

Title: Is nebulized saline a placebo in COPD?

Authors:

Shahina Y Khan (sykhan2001@yahoo.com)
B RONAN O'Driscoll (ronan.odriscoll@srht.nhs.uk)

Version: 4 Date: 29 September 2004

Author's response to reviews:

I have made all the changes exactly as requested in your e-mail dated 28th September 2004.
Title- underlining removed.
Qualifications removed.
Second author's e-mail address added to title page.
Keywords deleted.
Conflicts of interest renamed as "Competing interests" and moved to below Abbreviations.
Authors contributions changed to Authors' contributions.
All references now appear before full stop.
Tables now have cells with black borders as requested.
References now contain all authors' names.